Heritability and Genetic Correlations Explained by Common SNPs for Metabolic Syndrome Traits

Shashaank Vattikuti, Juen Guo, Carson C. Chow*

Laboratory of Biological Modeling, National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK), National Institutes of Health (NIH), Bethesda, Maryland, United States of America

Abstract

We used a bivariate (multivariate) linear mixed-effects model to estimate the narrow-sense heritability (h^2) and heritability explained by the common SNPs (h_g^2) for several metabolic syndrome (MetS) traits and the genetic correlation between pairs of traits for the Atherosclerosis Risk in Communities (ARIC) genome-wide association study (GWAS) population. MetS traits included body-mass index (BMI), waist-to-hip ratio (WHR), systolic blood pressure (SBP), fasting glucose (GLU), fasting insulin (INS), fasting trigylcerides (TG), and fasting high-density lipoprotein (HDL). We found the percentage of h^2 accounted for by common SNPs to be 58% of h^2 for height, 41% for BMI, 46% for WHR, 30% for GLU, 39% for INS, 34% for TG, 25% for HDL, and 80% for SBP. We confirmed prior reports for height and BMI using the ARIC population and independently in the Framingham Heart Study (FHS) population. We demonstrated that the multivariate model supported large genetic correlations between BMI and WHR and between TG and HDL. We also showed that the genetic correlations between the MetS traits are directly proportional to the phenotypic correlations.

Citation: Vattikuti S, Guo J, Chow CC (2012) Heritability and Genetic Correlations Explained by Common SNPs for Metabolic Syndrome Traits. PLoS Genet 8(3): e1002637. doi:10.1371/journal.pgen.1002637

Editor: Peter M. Visscher, The University of Queensland, Australia

Received July 22, 2011; Accepted February 21, 2012; Published March 29, 2012

This is an open-access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the Creative Commons CC0 public domain dedication.

Funding: This work was supported by the Intramural Research Program of the NIH/NIDDK. The Atherosclerosis Risk in Communities Study is carried out as a collaborative study supported by National Heart, Lung, and Blood Institute contracts N01-HC-55015, N01-HC-55016, N01-HC-55018, N01-HC-55019, N01-HC-55020, N01-HC-55022, R01HL087641, R01HL59367, and R01HL086694; National Human Genome Research Institute contract U01HG004402; and National Institutes of Health contract HNS268200625226C. Infrastructure was partly supported by Grant Number UL1RR025005, a component of the National Institutes of Health contract HHS026820625226C. Infrastructure was partly supported by Grant Number UL1RR025005, a component of the National Institutes of Health and NIH Roadmap for Medical Research. The authors thank the staff and participants of the ARIC study for their important contributions. The Framingham Heart Study is conducted and supported by the National Heart, Lung, and Blood Institute (NHLBI) in collaboration with Boston University (Contract No. N01-HC-25195). This manuscript was not prepared in collaboration with investigators of the Framingham Heart Study and does not necessarily reflect the opinions or views of the Framingham Heart Study, Boston University, or NHLBI. Funding for SHARe genotyping was provided by NHLBI Contract N02-HL-64278. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: carsonc@mail.nih.gov

Introduction

Obesity associated traits such as central adiposity, dyslipidemia, hypertension, and insulin resistance are major risk factors for type 2 diabetes and cardiovascular complications [1]. The constellation of these traits has been termed metabolic syndrome (MetS). Understanding the genetic factors underlying these traits and how they are correlated is clinically important. Large-scale genotyping investigations such as genome-wide association studies (GWAS) are useful tools for identifying genetic factors. However, significant genetic variants discovered in GWAS explain only a small proportion of the expected narrow-sense heritability, h^2 , defined as the ratio of additive genetic variance to phenotypic variance [2]. This discrepancy underlies the debate concerning "missing" genetic factors among the common variants [3,4].

The main approach of GWAS has been to identify significant single-nucleotide polymorphisms (SNPs) by examining each SNP individually for significance. The h^2 attributed to that marker is then given by $2f(1-f)a^2$, where f is the frequency of the marker and a is the additive effect. To reduce the chance of false positives, a stringent p-value criterion has been adopted (typically $p = 5*10^{-8}$, based on an adjusted p-value of 0.05 for one-million tests). It has been suggested that this selection criterion is too conservative [5] and that some of the missing heritability may be

linked to genetic markers of small effect that fail this stringent cutoff.

Alternatively, the narrow sense heritability explained by the common SNPs, h_g^2 , may be estimated by adapting a linear mixedeffects model [6,7] that is used to estimate h^2 . This model decomposes the phenotypic variance into genetic and residual variance components. Usually, the model is applied to related individuals where the genetic relationships are estimated by using family pedigree or genetic markers [8,9]. Yang et al. [6,7] pointed out that h_{σ}^{2} could be estimated using genetic relationships obtained from the common SNPs for unrelated individuals. The main assumed difference between h_g^2 and h^2 is due to the difference in linkage disequilibrium (LD) between the common SNP markers and the rest of the genome, with the assumption that closely related individuals would be in greater LD than unrelated individuals. Thus, heritability estimated with the genetic relationships of unrelated individuals is attributed to the common variants while that estimated with genetic relationships of related individuals is attributed to the entire genome. While the method does not identify single variants, it provides the maximum expected variance expected by the set of markers or the relative complement of the set (e.g., common versus rare variants). Recently, it has been shown that a large proportion of h^2 is explained by the common single-nucleotide polymorphisms (SNPs)

Author Summary

The narrow-sense heritability of a trait such as body-mass index is a measure of the variability of the trait between people that is accounted for by their additive genetic differences. Knowledge of these genetic differences provides insight into biological mechanisms and hence treatments for diseases. Genome-wide association studies (GWAS) survey a large set of genetic markers common to the population. They have identified several single markers that are associated with traits and diseases. However, these markers do not seem to account for all of the known narrow-sense heritability. Here we used a recently developed model to quantify the genetic information contained in GWAS for single traits and shared between traits. We specifically investigated metabolic syndrome traits that are associated with type 2 diabetes and heart disease, and we found that for the majority of these traits much of the previously unaccounted for heritability is contained within common markers surveyed in GWAS. We also computed the genetic correlation between traits, which is a measure of the genetic components shared by traits. We found that the genetic correlation between these traits could be predicted from their phenotypic correlation.

for several traits using this model [6,7]. Here, we showed that large proportions of the phenotypic variance for several metabolic syndrome (MetS) traits were also captured by the common SNPs. Among these, we validated the height and body-mass index estimates by Yang et al. [6,7] in independent GWAS populations. We also quantified the genetic correlation between traits explained by the common SNPs.

Results

We estimated h^2 and h_g^2 for height and body-mass index (BMI) in the Framingham Heart Study population (FHS), and height and seven metabolic syndrome traits (MetS) traits: BMI, waist-to-hip ratio (WHR), systolic blood pressure (SBP), fasting glucose (GLU), fasting insulin (INS), fasting triglycerides (TG), and fasting highdensity lipoprotein (HDL) in the Atherosclerosis Risk in Communities population (ARIC) (ARIC MetS estimates shown in Table 1). Our base FHS population consisted of 4,240 subjects and our base ARIC population consisted of 8,451 subjects (see Methods and Tables S1 and S2 for a description of the populations). The genetic relationship between pairs of subjects was estimated using 436,126 genome-wide common SNP markers for ARIC and 320,118 SNPs for FHS (see Methods for details).

We first estimated h^2 for related individuals with relationships between 0.35 and 0.65, derived empirically from the SNP markers, for height and BMI in the ARIC and FHS populations (see Methods for derivation of the relationship matrix). This resulted in 3,663 subjects (6,706,953 pairs of subjects) for FHS and 530 subjects (140,185 pairs of subjects) for ARIC. We found h^2 to be 0.77 (s.e. 0.03) for height and 0.39 (s.e. 0.04) for BMI in FHS, and 0.88 (s.e. 0.09) for height and 0.34 (s.e. 0.12) for BMI in ARIC. The estimated h^2 were consistent with values obtained using phenotypic regression (data not shown) and previous results [6,7,10,11].

We then compared these values to estimates for h_{ρ}^{2} for unrelated individuals with relationships less than 0.025 (see Methods for derivation of the relationship matrix). This resulted in 1,489 subjects (1,107,816 pairs of subjects) for FHS and 5,647 subjects (31,882,962 pairs of subjects) for ARIC. As mentioned above, h_a^2 provides an estimate of the heritability explained by common variants because of presumed lesser linkage disequilibrium between the common SNPs and the rest of the genome as compared to related individuals. We found h_{g}^{2} to be 0.50 (s.e. 0.18) for height and 0.10 (s.e. 0.18) for BMI in FHS, and 0.46 (s.e. 0.05) for height and 0.14 (s.e. 0.05) for BMI in ARIC. These values are consistent with previously estimated values [6,7]. Using the average across FHS and ARIC estimates, this implied that the common SNPs accounted for approximately 58% of h^2 for height and 33% for BMI. To assess whether including more common SNPs would explain more of the h^2 , we examined how h_g^2 depended on the number of SNPs. As shown in Figure S1, the mean and standard error of the h_{g}^{2} estimate for height in the ARIC population appeared to stabilize after approximately 300,000 SNPs.

We then estimated h^2 and h_g^2 for the MetS traits in the ARIC population using the same subjects as above (see Table 1). We validated our h^2 estimates by using phenotypic regression between related individuals for some of the traits (data not shown). The median h^2 was 0.33, the minimum was 0.23 (INS), and the maximum was 0.48 (HDL). The median h_g^2 was 0.13, the minimum was 0.09 (INS), and maximum was 0.24 (SBP). Comparing the medians suggested that h_g^2 explains ~39% of the h^2 for these MetS traits. We found that the common SNPs explained large proportions of the h^2 : 41% of h^2 for BMI, 46% for WHR, 30% for GLU, 39% for INS, 34% for TG, 25% for HDL, and 80% for SBP.

We next estimated the genetic correlations between MetS traits using a bivariate (multivariate) model (see Tables S3 and S4 for covariances). Table 2 shows the genetic and residual correlations for related individuals using bivariate models. The genetic correlation is the additive genetic covariance between traits normalized by the geometric mean of the individual trait genetic variances. The residual correlation is similarly estimated using the residual covariance and variances. For related individuals, we found significant genetic correlations for BMI-WHR, WHR-INS, GLU-INS, INS-TG, and TG-HDL and significant residual correlations between BMI-WHR, BMI-INS, BMI-HDL, WHR-INS, INS-HDL, and TG-HDL. Table 3 shows the genetic and residual correlations for the unrelated individuals. We found significant genetic correlations for BMI-WHR and TG-HDL and significant residual correlations for all of the estimates except SBP-

Table 1. h^2 and h_g^2 estimates (ARIC population).

	BMI	WHR	GLU	INS	TG	HDL	SBP
h²	0.34 (0.12)	0.28 (0.12)	0.33 (0.12)	0.23 (0.12)	0.47 (0.11)	0.48 (0.11)	0.30 (0.12)
h _g ²	0.14 (0.05)	0.13 (0.05)	0.10 (0.05)	0.09 (0.05)	0.16 (0.05)	0.12 (0.05)	0.24 (0.05)

Mean and standard error estimates from univariate models. doi:10.1371/journal.pgen.1002637.t001
 Table 2. Genetic and residual correlation coefficients between MetS traits in the ARIC population among related individuals from

 the bivariate REML model.

	ВМІ	WHR	GLU	INS	TG	HDL	SBP
BMI		0.75 (0.16)*	0.23 (0.24)	0.17 (0.27)	0.19 (0.20)	-0.12 (0.21)	0.55 (0.24)
WHR	0.52 (0.08)*		0.35 (0.26)	0.67 (0.26)*	0.10 (0.22)	-0.12 (0.22)	0.37 (0.26)
GLU	0.19 (0.12)	0.14 (0.12)		0.69 (0.25)*	0.21 (0.21)	-0.07 (0.21)	0.13 (0.27)
INS	0.64 (0.08)*	0.35 (0.09)*	0.22 (0.11)		0.76 (0.21)*	-0.33 (0.23)	0.29 (0.29)
TG	0.29 (0.12)	0.34 (0.12)	0.21 (0.13)	0.27 (0.11)		-0.59 (0.13)*	0.21 (0.22)
HDL	-0.38 (0.12)*	-0.34 (0.12)	-0.22 (0.13)	-0.39 (0.11)*	-0.45 (0.11)*		-0.06 (0.23)
SBP	0.11 (0.12)	0.18 (0.11)	0.05 (0.12)	0.24 (0.11)	0.10 (0.13)	-0.02 (0.13)	

Mean and standard error of the Pearson correlation coefficient for genetic correlations (upper triangle) and residual correlations (lower triangle). An asterisk indicates significance with p < 0.05 adjusted for 21 hypotheses using the two-tailed hypothesis test and normal distribution of the Fisher transformed correlation coefficient. doi:10.1371/journal.pgen.1002637.t002

HDL. The genetic correlations for unrelated individuals were proportional to the genetic correlations for related individuals (see Figure S2) with a proportionality constant of 0.44 (s.e. = 0.15; two-tail *t*-distribution *p*-value with 20 d.f. = $8.2*10^{-3}$). The phenotypic correlations between traits were similar for related and unrelated individuals and are shown in Table 4. These values were also consistent with the reported estimates in the National Heart Lung and Blood Institute-Family Heart Study (NHLBI-FHS), which included Framingham Heart Study and ARIC families [11].

We validated our genetic correlation estimates using bivariate models for each pair of traits by analyzing all 7 MetS traits simultaneously for the unrelated individuals in a single multivariate model. This 7 trait multivariate model was much more expensive computationally so we used a less stringent convergence rule. The results were similar to the bivariate model (see Table S5 and S6) although the genetic correlation increased and their error decreased for a number of the estimates. In addition to the significant genetic correlations in the bivariate models, we also found the genetic correlation for BMI-INS to be significant in the 7 trait model.

We then examined the relationship between the genetic and phenotypic correlations (see Figure S3). For related individuals, we found that the phenotypic correlations r_p were proportional to the genetic correlations r_g with a proportionality constant of 1.2 (s.e. = 0.16; two-tail *t*-distribution *p*-value with 20 d.f. = $3.1*10^{-7}$). For unrelated individuals, we found that the phenotypic correlations were proportional to the genetic correlations with a

proportionality constant of 0.85 (s.e. = 0.19; two-tail *t*-distribution *p*-value with 20 d.f. = $2.3*10^{-4}$). The direct proportionality between r_p and r_g implies that the ratio r_g/r_p is approximately constant for the MetS traits.

Discussion

We used a recently developed approach to analyzing GWAS data and provided new estimates for the total amount of additive genetic information contained in the common SNPs for MetS traits. The approach uses a linear mixed-effects model to estimate the additive genetic variances and correlations between traits. The model relies on knowing the genetic relationships between the individuals analyzed. Previously, this had been obtained from family pedigrees. Visscher et al. [9] and Yang et al. [6] observed that the genetic relationships could be computed from the GWAS SNPs. They also presumed that the heritability estimated for unrelated individuals with low SNP correlation are explained mainly by these common SNPs because the linkage disequilibrium between the common SNPs and the rest of the genome is weak. This would be in contrast to related individuals with high SNP correlation where linkage disequilibrium is strong. Thus, heritability estimated with the genetic relationships of unrelated individuals is attributed to the common SNPs while that estimated with the related individuals is attributed to the entire genome. This then creates a major distinction between h^2 and h_g^2 . We computed both in the same population. However, differences between

Table 3. Genetic and residual correlations between MetS traits in the ARIC population among unrelated individuals from the bivariate REML model.

	ВМІ	WHR	GLU	INS	ТG	HDL	SBP
BMI		0.91 (0.18)*	0.01 (0.32)	0.57 (0.24)	0.20 (0.24)	-0.15 (0.28)	0.16 (0.20)
WHR	0.44 (0.03)*		0.09 (0.32)	0.33 (0.31)	0.32 (0.23)	-0.06 (0.30)	0.17 (0.21)
GLU	0.27 (0.04)*	0.18 (0.04)*		0.05 (0.40)	0.07 (0.30)	-0.16 (0.34)	0.11 (0.24)
INS	0.51 (0.03)*	0.40 (0.04)*	0.39 (0.04)*		0.22 (0.29)	-0.20 (0.36)	0.20 (0.25)
TG	0.31 (0.04)*	0.33 (0.04)*	0.20 (0.04)*	0.43 (0.04)*		-0.57 (0.19)*	0.002 (0.19)
HDL	-0.34 (0.04)*	-0.33 (0.04)*	-0.16 (0.04)*	-0.39 (0.04)*	-0.51 (0.03)*		-0.03 (0.22)
SBP	0.25 (0.05)*	0.18 (0.05)*	0.17 (0.05)*	0.22 (0.04)*	0.21 (0.05)*	-0.04 (0.05)	

Mean and standard error of the Pearson correlation coefficient for genetic correlations (upper triangle) and residual correlations (lower triangle). An asterisk indicates significance with p < 0.05 adjusted for 21 hypotheses using the two-tailed hypothesis test and normal distribution of the Fisher transformed correlation coefficient. doi:10.1371/journal.pgen.1002637.t003

Table 4. Phenotypic correlation coefficients between MetS traits in the ARIC population.

	ВМІ	WHR	GLU	INS	тG	HDL	SBP
BMI		0.59 (0.04)*	0.20 (0.04)*	0.49 (0.04)*	0.24 (0.04)*	-0.26 (0.04)*	0.25 (0.04)*
WHR	0.51 (0.01)*		0.21 (0.04)*	0.43 (0.04)*	0.23 (0.04)*	-0.24 (0.04)*	0.23 (0.04)*
GLU	0.24 (0.01)*	0.17 (0.01)*		0.34 (0.04)*	0.21 (0.04)*	-0.15 (0.04)*	0.07 (0.04)
INS	0.52 (0.01)*	0.39 (0.01)*	0.35 (0.01)*		0.42 (0.04)*	-0.35 (0.04)*	0.25 (0.04)*
TG	0.30 (0.01)*	0.33 (0.01)*	0.19 (0.01)*	0.40 (0.01)*		-0.52 (0.04)*	0.14 (0.04)*
HDL	-0.32 (0.01)*	-0.30 (0.01)*	-0.15 (0.01)*	-0.37 (0.01)*	-0.52 (0.01)*		-0.04 (0.04)
SBP	0.23 (0.01)*	0.18 (0.01)*	0.15 (0.01)*	0.21 (0.01)*	0.16 (0.01)*	-0.04 (0.01)*	

Mean and standard error of the Pearson correlation coefficient. Coefficients among related individuals shown in the upper triangle. Coefficients among unrelated individuals shown in the lower triangle. An asterisk indicates significance with p < 0.05 adjusted for 21 hypotheses using the two-tailed hypothesis test and normal distribution of the Fisher transformed correlation coefficient.

doi:10.1371/journal.pgen.1002637.t004

estimates of h^2 and h_g^2 may also arise due to differences in environmental influences and non-additive genetic effects that may bias the estimates. Provided that these biases are small then the ratio of h_g^2 to h^2 provides an estimate of the proportion of narrow sense heritability captured by the common SNPs.

We confirmed previous findings that a large proportion of h^2 is explained by the common SNPs. Our h_{ϱ}^{2} estimates for height and BMI in two independent analyses (i.e. ARIC and FHS) were consistent with previously reported values [6,7]. Our h^2 estimates for BMI, GLU, INS, TG, HDL, and SBP were similar to the findings of the large family National Heart, Lung, and Blood Institute (NHLBI) Family Heart Study [11], which included Framingham Heart Study and ARIC families. We found that h_g^2 explained a large proportion of h^2 across the MetS traits, and h_g explained approximately 39% of the h^2 for these traits. We estimated that the common SNPs explain 58% of h^2 for height, 41% for BMI, 46% for WHR, 30% for GLU, 39% for INS, 34% for TG, 25% for HDL, and 80% for SBP. Our h_{σ}^{2} findings are striking compared to traditional GWAS approaches where significant common SNPs have been shown to explain only 4% of h^2 for BMI with 32 SNPs, 11% for GLU with 14 SNPs, 20% for TG with 48 SNPs, 25% for HDL with 60 SNPs, 3% for SBP with 10 SNPs, and 12% for height with 180 SNPs [12-16]. Height had the largest absolute h_g^2 , which was consistent with having a large h^2 . Surprisingly, SBP had the largest proportion of h^2 explained by the common SNPs while only a few percent of this has been uncovered by traditional GWAS. However, the standard error of h_{ρ}^{2} for SBP was large and reducing this error will be important for further investigation. Conversely, our analysis suggested that the SNP markers already identified for TG and HDL may contain the maximum heritability expected from the common SNPs.

Our analysis of h_g^2 against the number of SNPs suggested that the mean and standard error of h_g^2 for height is well estimated by approximately 300,000 markers and that including more markers would have little effect for this trait and perhaps others. The standard error of h_g^2 also increased with SNP number. This may seem paradoxical but can be explained by recalling that the estimate for h_g^2 is proportional to the regression coefficient of the square of the phenotype differences versus the genetic relationship (i.e. Haseman-Elston regression) [8]. The standard error of h_g^2 is thus inversely proportional to the variance of the genetic relationship. Since the latter is estimated from the common SNPs, this variance is expected to decrease as the number of SNPs increases thereby increasing the standard error [6].

Using the bivariate (multivariate) model [17,18] we estimated the genetic and residual correlations between the MetS traits. Among these, we found that the genetic correlations in related and unrelated individuals for BMI and WHR were significantly different from zero. This is consistent with both traits as indirect measures of body fat and common health risks [19]. Previously, Rice et al., 1994 [20] found significant genetic correlations between BMI and SBP among normotensive nonobese families. This suggested a common genetic etiology to their physiological relationship through hyperinsulinemia resulting in increased renal reabsorption of sodium and sympathetic activation [20]. We found a large genetic correlation among related subjects, although it was not significant because of the large error. This was consistent with the large family study by the NHLBI that did not find a significant genetic correlation [8]. Perusse et al, 1997 [21] argued that crosstrait resemblance between BMI and lipids is mostly environmental. In concordance, we did not find significant genetic correlations between either BMI or WHR and TG and HDL for either related or unrelated individuals (see Table 3 and Table 4) while residual (which includes environmental) correlations were significant for BMI-HDL. We found that the residual covariance accounted for a minimum of 71% (derived from the estimates in Table 4 and Table S3) of the phenotype covariance between BMI or WHR and the lipid measurements for related individuals. Genetic correlations between TG and HDL were also large, which is consistent with their direct physiological relationship [22]. This is also consistent with the findings from a recent GWAS meta-analysis whose results showed that 50% of the significant markers for TG were also significant for HDL (derived from Supplementary Tables 6 and 11 in [16]), and with a genome-wide LOD correlation analysis [23]. While we found some significant genetic correlations among both related and unrelated subjects, the variance was large for these estimates and greater statistical power is needed for better accuracy.

We found that the genetic correlation was directly proportional to the phenotypic correlation, which was an unexpected, empirical finding. Previously, a linear relationship between the correlations was hypothesized by Cheverud for sets of traits with common functions, and shown empirically for a number of traits [8,24–26]. While this finding is interesting from an evolutionary genetics perspective, it may also serve a useful purpose in the maximum likelihood computation of the linear mixed-effects model by providing initial genetic correlation (i.e. covariance) estimates based on the phenotypic correlations.

In summary, we provided evidence that the common SNPs explain large proportions of the variance for several MetS traits in agreement with previous findings for some of these traits [6,7]. This is consistent with the original premise of GWAS that a large

proportion of phenotypic variation for common traits may be due to common variants [27]. However, an amendment to this premise is that it is likely to be many common variants with small effect. This is supported by recent meta-analyses with larger sample sizes that have identified more associated common SNPs. This approach can serve as a first approximation of the total heritability expected from common SNPs given a genome-wide set of markers and requires fewer subjects to achieve significant results. We also found genetic associations that will be useful for single gene and systems biology studies. Future studies with greater power will provide estimates for weaker multivariate genetic associations and provide greater precision for the estimates presented here.

Methods

ARIC population and GWAS data

Our main study population was the Atherosclerosis Risk In Communities (ARIC) population. The ARIC population consists of a large sample of unrelated individuals and some families across North America. The population was recruited from four centers across the United States: Forsyth County, North Carolina; Jackson, Mississippi; Minneapolis, Minnesota; and Washington County, Maryland. For this study, we restricted our analysis to the European-American group. The population was recruited in 1987 from the general population consisting of subjects aged 45 to 64 years. The ARIC population consisted of 8,451 subjects.

Quality control and genotype calls for common SNPs were evaluated previously for ARIC using the Affymetrix Human SNP Array 6.0. We selected bilallelic autosomal markers based on the following criteria: missingness <0.05, Hardy-Weinberg equilibrium $(p < 10^{-6})$ and minor allele frequency >0.05. Subjects with missingness >0.05 were removed. This resulted in 436,126 retained markers.

Ouality control measurements from dbGAP (GENEVA ARIC Project Quality Control Report Sept 22, 2009) indicate significant population stratification between self-identified white (Europeanancestory kind group) and black populations when projected onto HapMap components. Furthermore, principal-components analysis of the European-ancestory group by dbGAP showed that no component explained more than 0.1% of the population variance. For this study we only analyzed the European-ancestory group and treated it as a single population.

ARIC phenotypes were adjusted for age, sex, and study center. Only single measurements from visit 1 were used for these subjects. We only used subjects with negative diabetes status and with genotype and phenotype information for all traits. This resulted in 8,451 subjects. We standardized all the traits. We first log-transformed BMI, glucose, insulin, triglycerides, HDL, and systolic blood pressure. All laboratory measurements are under fasting conditions. Population trait statistics are in Table S1.

Framingham Heart Study (FHS) population

We estimated h^2 and h_q^2 for height and BMI in the Framingham Heart Study population (FHS). The FHS population is a large multi-generational dataset that started in 1948 in Framingham, Massachusetts in the United States. It consists of a number of ethnicities predominantly from the United Kingdom, Ireland, Italy, and Western Europe [28]. Markers were screened similarly to ARIC and we also removed any SNPs that did not overlap with the ARIC set, which results in 320,118 SNPs. We used principal components analysis of the linkage disequilibrium (LD) pruned genetic relationship matrix to identify components with variance >0.1%. LD pruning was as in the ARIC 2009 report. This resulted in 73,432 retained SNPs. We found three significant components that were then used as covariates in the REML model. For consistency with ARIC, we restricted the age range at time of exam to 45 to 65 years and randomly selected a single measurement in the case of multiple measurements. Phenotypes were adjusted for age, sex, and generation prior to the REML estimation and standardized. We first log-transformed BMI. Population trait statistics are in Table S2. Our base FHS population consisted of 4,240 subjects.

h^2 estimates using common SNP estimated relationship

We determined h^2 using the linear mixed-effects model (see derivation below) and related individuals defined as genomic relatedness between 0.35 and 0.65. We assume that the common SNPs are in greater linkage disequilbrium among related individuals and, as such, can be used to estimate the total additive-genetic variance across the allele spectrum as suggested by Visscher et al., 2006 [9]. We constrained the relationship matrix to have at least one related pair per subject. This was done by pruning the entire population relationship matrix by randomly selecting a row and removing the row and its corresponding column if no genomic covariance in the row was between the cutoff values. For all pairs, including unrelated individuals, we used their empirically defined relationship. This resulted in 530 individuals being selected for analysis in ARIC and 3,663 individuals in FHS.

 h^2 was estimated with $h^2 = var_e/(var_e + var_e)$, where var_e and var_e are the genetic and residual variance components estimated by the REML model using related individuals. The error was estimated from the inverse Fisher Information (see linear mixed-effects model below) and propagated using a first-order Taylor expansion.

Common SNP linear mixed-effects model estimate of h_q^2

We used the linear mixed-effects model and only unrelated individuals to estimate the additive-genetic variance attributable to the common SNPs (h_g^2) . Unrelated individuals were defined as subjects with maximum genomic correlation of <0.025. The genomic relationship matrix was then produced as above based on this cutoff. The cutoff was taken from Yang et al. 2010 [6] and is less than the expected coefficient of relatedness between 2nd cousins. For these estimates we used the same group of 5,647 unrelated individuals for all estimates in ARIC and 1,489 individuals in FHS. h_g^2 was estimated as $h_g^2 = var_g/(var_g + var_e)$, where var_{a} and var_{e} are the genetic and residual variance components estimated by the REML model using unrelated individuals. The standard error was estimated as above. The height hg² versus SNP number analyses were performed over allele frequency range of 0.05 to 0.5 in order of increasing and decreasing frequency.

Correlations

The genetic correlation (r_g) is defined as $r_g = \frac{\text{cov}_g(t_1, t_2)}{\sqrt{\text{var}_g(t_1)\text{var}_g(t_2)}}$ where $(var_g(t_i))$ is the additive genetic variance of trait *i* and covariance $(cov_g(t_i, t_j))$ is the additive genetic covariance between the traits. The variances and covariances are estimated directly in the multivariate linear mixed-effects model. The error was computed from the estimated errors of the variances and covariance using a first-order Taylor expansion. The residual and phenotypic correlations were analogously defined. Phenotype correlations and error were estimated by linear regression of the standardized phenotypes.

Proportionality constants

The mean and errors for proportionality constants between the genetic and phenotypic correlations were determined by randomly sampling over the distributions of the parameter estimates (i.e. Monte Carlo method) assuming that the error around the mean parameter estimate was normally distributed and that the parameters were independent. We then fit a linear function with the y-intercept fixed at 0 (after first confirming that it was not significantly different from zero).

Significance testing

We assessed significance for correlation coefficients (*r*) using the standardized Fisher transformed estimate of r: $\arctan(r)/\arctan(s.e.(r))$. We estimated the two-tailed p-value from a normal distribution and significance was determined by p<0.05 and Bonferroni corrected for 21 hypotheses.

Significance for regression coefficient (β) was estimated using the standardized coefficient $\beta/s.e.(\beta)$. We estimated the two-tailed p-value from a t-distribution and 20 degrees of freedom and significance was determined by p<0.05.

Preprocessing of SNPs and phenotypes was done using PLINK [29] (v1.07,http://pngu.mgh.harvard.edu/purcell/plink/) and MATLAB (2010b, MathWorks, Natick, MA). REML optimization was executed using software written in MATLAB.

Bivariate (multivariate) linear mixed-effects linear model

We considered the following multivariate linear mixed-effects model for m individuals, n loci and t traits [6–8,17,18,30]:

$$\mathbf{y}_i = \mathbf{X}_i \mathbf{v}_i + \mathbf{Z} \mathbf{u}_i + \mathbf{e}_i$$

where \mathbf{y}_i is a $m \times 1$ vector of trait i for m individuals, \mathbf{X}_i is an $m \times s$ fixed effects matrix for trait i, \mathbf{v}_i is a $s \times 1$ vector of fixed effects parameters for trait i, \mathbf{Z} is an $m \times n$ matrix of standardized genotypes, \mathbf{u}_i is an $n \times 1$ vector of random effects for trait i satisfying $\mathbf{u}_i \sim \mathbf{N}(\mathbf{0}, \mathbf{G})$ and \mathbf{e}_i is an $m \times 1$ vector of residual effects satisfying $\mathbf{e}_i \sim \mathbf{N}(\mathbf{0}, \mathbf{G})$, with matrix blocks $\mathbf{G}_{ij} = \operatorname{cov}_{gij} \mathbf{I}_n$ and $\mathbf{R}_{ij} = \operatorname{cov}_{gij} \mathbf{I}_n$ and \mathbf{I}_l is the $l \times l$ identity matrix. This model can be used for single or multiple traits. For two traits, it is called a bivariate model. The model is identical to that used by [6,7,17].

We considered only bi-allelic SNPs in Hardy-Weinberg equilibrium. Denote the minor allele by q and the major allele by Q. Let the minor allele frequency at locus *i* have frequency p_i . We assign a value of 2 for genotype qq, 1 for genotype qQ and 0 for genotype QQ. The Hardy-Weinberg mean frequency for the genotype at locus *i* is $2p_i$ and the variance is $2p_i(1-p_i)$. The standardized genotype entries have values of $(2-2p_i)/(2p_i(1-2p_i))^{1/2}$ for qq, $(1-2p_i)/(2p_i(1-2p_i))^{1/2}$ for qQ genotype.

The log of the likelihood function is given by

$$L = -\frac{n}{2}ln(2\pi) - \frac{1}{2}ln|\mathbf{V}| - \sum_{i=1,j=1}^{l} (\mathbf{y}_i - \mathbf{X}_i \mathbf{v}_i) \mathbf{V}_{ij}^{-1} (\mathbf{y}_j - \mathbf{X}_j)$$

where the covariance matrix can be expressed as a tensor product $\mathbf{V} = \mathbf{G} \otimes \mathbf{A} + \mathbf{R} \otimes \mathbf{I}$ with $m \times m$ blocks \mathbf{V}^{-1}_{ij} and \mathbf{A} is the genetic relationship matrix. Following Yang et al. [6], we used a modified covariance matrix for \mathbf{A} , $\mathbf{A} = \mathbf{Z}\mathbf{Z}'/n$, where the diagonals of \mathbf{A} are computed using the formula

$$A_{kl} = \frac{1}{n} \sum_{k=1}^{m} Z_{kl} \left(Z_{kl} + \frac{2p_l - 1}{\sqrt{2p_l(1 - p_l)}} \right)$$

We use the restricted maximum likelihood (REML) approach [8] where the gradients of the log likelihood are given by

$$\frac{\partial L}{\partial cov_{gij}} = \frac{1}{2} \mathbf{y}' \mathbf{P} \mathbf{A}_{ij} \mathbf{P} \mathbf{y} - \frac{1}{2} Tr(\mathbf{P} \mathbf{A}_{ij})$$
$$\frac{\partial L}{\partial cov_{eij}} = \frac{1}{2} \mathbf{y}' \mathbf{P} \mathbf{I}_{ij} \mathbf{P} \mathbf{y} - \frac{1}{2} Tr(\mathbf{P} \mathbf{I}_{ij})$$

where \mathbf{I}_{ij} is a $tm \times tm$ dimensional matrix with zero entries except for a $m \times m$ identity matrix at block location *i*, *j*, $\mathbf{A}_{ij} = \mathbf{A} \otimes \mathbf{I}_{ij}$ and $\mathbf{P} = \mathbf{V}^{-1} - \mathbf{V}^{-1} \mathbf{Y} (\mathbf{Y}' \mathbf{V}^{-1} \mathbf{Y})^{-1} \mathbf{Y}' \mathbf{V}^{-1}$, where $\mathbf{Y} = \mathbf{I}_t \otimes \mathbf{X}_i$.

We solved the REML equations using an EM algorithm [8], which was given by

$$(cov_{gij})^{k+1} = (cov_{gij})^k + \frac{(cov_{gij})^k}{m} \left(\mathbf{y}' \mathbf{P}^k \mathbf{A}_{ij} \mathbf{P}^k \mathbf{y} - Tr(\mathbf{P}^k \mathbf{A}_{ij}) \right)$$

$$(cov_{eij})^{k+1} = (cov_{eij})^k + \frac{(cov_{eij})^k}{m} \left(\mathbf{y}' \mathbf{P}^k \mathbf{I}_{ij} \mathbf{P}^k \mathbf{y} - Tr(\mathbf{P}^k \mathbf{I}_{ij}) \right)$$

for iteration k+1 in terms of iteration k. We iterated until the rate of change of the log likelihood function was less than about 10^{-4} . We also checked that the rate of change of the square of the covariance predictions was less than 10^{-8} . We checked our results against the software developed by Yang et al. (GCTA) [31] for the univariate model.

For the multivariate model, we transformed to a coordinate system where the covariance matrices were diagonal [8] to speed up the computation. Let \mathbf{z}_j be the set of phenotypes for individual *j*. We used the canonical transformation $\tilde{\mathbf{z}}_j = \mathbf{Q}\mathbf{z}_j$ such that $\mathbf{Q}\mathbf{G}\mathbf{Q}' = \mathbf{\Lambda}$ and $\mathbf{Q}\mathbf{R}\mathbf{Q}' = \mathbf{I}_t$. \mathbf{Q} can be computed from the formula $\mathbf{Q} = \sqrt{\mathbf{S}\mathbf{R}\mathbf{S}'\mathbf{S}^{-1}}$ where $\mathbf{S}\mathbf{G}\mathbf{R}^{-1} = \mathbf{\Lambda}\mathbf{S}$, (\mathbf{S} is the matrix of left eigenvectors of $\mathbf{G}\mathbf{R}^{-1}$). The transformed genetic covariances are given by $\boldsymbol{\Lambda}$ and the residual covariances are \mathbf{I}_t . Each step consisted of taking a single step with the univariate EM algorithm for the transformed additive genetic and residual variance followed by a transformation back to the original coordinates. We iterated until the maximum of the magnitudes of the components of the gradient of the log likelihood function was less than approximately 5×10^{-4} .

In our computations, we used both the direct EM algorithm and the canonically transformed algorithm because even though the transformed algorithm was in principle faster, it sometimes had poor convergence properties if the initial guess was not sufficiently close to the maximum likelihood value. We ensured that both give the same results. For computational efficiency, the results shown are computed from the bivariate model for the different trait pairs. We confirmed our results with a multivariate model that included all traits.

Our error estimates were given by the inverse of the Fisher information matrix **F**, which we computed by evaluating the Hessian of the log likelihood at the maximum likelihood predictions. **F** is a $t(t+1) \times t(t+1)$ dimensional matrix with rows corresponding to the genetic and residual variances and covariances (where cov_{ij} was set equal to cov_{ji}) and with block elements (that are not all contiguous) given by

$$\mathbf{F}_{ij,kl} = \frac{1}{2} \begin{pmatrix} Tr(\mathbf{P}\mathbf{A}_{ij}\mathbf{P}\mathbf{A}_{kl}) & Tr(\mathbf{P}\mathbf{A}_{ij}\mathbf{P}\mathbf{A}_{kl}) \\ Tr(\mathbf{P}\mathbf{A}_{ij}\mathbf{P}\mathbf{I}_{kl}) & Tr(\mathbf{P}\mathbf{I}_{ij}\mathbf{P}\mathbf{I}_{kl}) \end{pmatrix}$$

for $i \leq j$ and $k \leq l$.

Supporting Information

Figure S1 Height h_g^2 versus number of SNPs by sampling the allele frequency from 0.05 to 0.5 (red = low to high, blue = high to low, black = using all SNPs). A) h_g^2 estimates for height relative to the number of SNPs (mean and s.e.). B) Standard error versus number of SNPs.

(TIF)

Figure S2 Genetic correlation coefficient for unrelated individuals versus the genetic correlation coefficients for related individuals. Shown are the mean and standard errors. Dashed line is the least squares fit with the y-intercept fixed at 0 estimated using a Monte Carlo method (slope = 0.44).

(TIF)

Figure S3 A) Genetic correlation coefficients versus the phenotypic correlation coefficients for related individuals. Shown are the mean and standard errors. Dashed line is the least squares fit with the y-intercept fixed at 0 estimated using a Monte Carlo method (slope = 1.2). B) Genetic correlation coefficients versus the phenotype correlation coefficients for unrelated individuals. Shown are the mean and standard errors. Dashed line is the least squares fit with the y-intercept fixed at 0 estimated using a Monte Carlo method (slope = 0.85).

(TIF)

Table S1 Atherosclerosis Risk in Communities Study (ARIC) population statistics by sex; mean (sd; minimum-maximum). BMI = body-mass index, WC = waist circumference, WHR = wa-ist-to-hip ratio, GLU = fasting glucose, INS = fasting insulin, TG = fasting triglycerides, HDL = fasting high-density lipoprotein, SBP = systolic blood pressure.

(DOCX)

 Table S2
 Framingham Heart Study (FHS) population statistics.

 (DOCX)
 (DOCX)

Table S3 Genetic and residual covariance estimates for the ARIC population among related individuals. Mean and standard

References

- Permutt MA, Wasson J, Cox N (2005) Genetic epidemiology of diabetes. J Clin Invest 115: 1431–1439.
- Visscher PM, Hill WG, Wray NR (2008) Heritability in the genomics eraconcepts and misconceptions. Nat Rev Genet 9: 255–266.
- 3. Gibson G (2010) Hints of hidden heritability in GWAS. Nat Genet 42: 558–560.
- Maher B (2008) Personal genomes: The case of the missing heritability. Nature 456: 18–21.
- Pearson TA, Manolio TA (2008) How to interpret a genome-wide association study. JAMA 299: 1335–1344.
- Yang J, Benyamin B, McEvoy BP, Gordon S, Henders AK, et al. (2010) Common SNPs explain a large proportion of the heritability for human height. Nat Genet 42: 565–569.
- Yang J, Manolio TA, Pasquale LR, Boerwinkle E, Caporaso N, et al. (2011) Genome partitioning of genetic variation for complex traits using common SNPs. Nat Genet 43: 519–525.
- Lynch M, Walsh B (1998) Genetics and analysis of quantitative traits. Sunderland, Mass.: Sinauer. xvi, 980 p.
- Visscher PM, Medland SE, Ferreira MA, Morley KI, Zhu G, et al. (2006) Assumption-free estimation of heritability from genome-wide identity-by-descent sharing between full siblings. PLoS Genet 2: e41. doi:10.1371/journal.pgen. 0020041.
- Coady SA, Jaquish CE, Fabsitz RR, Larson MG, Cupples LA, et al. (2002) Genetic variability of adult body mass index: a longitudinal assessment in framingham families. Obes Res 10: 675–681.
- Tang W, Hong Y, Province MA, Rich SS, Hopkins PN, et al. (2006) Familial clustering for features of the metabolic syndrome. Diabetes Care 29: 631.
- Dupuis J, Langenberg C, Prokopenko I, Saxena R, Soranzo N, et al. (2010) New genetic loci implicated in fasting glucose homeostasis and their impact on type 2 diabetes risk. Nat Genet 42: 105–116.
- Lango Allen H, Estrada K, Lettre G, Berndt SI, Weedon MN, et al. (2010) Hundreds of variants clustered in genomic loci and biological pathways affect human height. Nature 467: 832–838.

error of genetic (upper triangle) and residual (lower triangle) covariance estimates from the univariate (diagonals) and bivariate (off-diagonals) REML model. (DOCX)

Table S4 Genetic and residual covariance estimates for the ARIC population among unrelated individuals. Mean and standard error of genetic (upper triangle) and residual (lower triangle) covariance estimates from the univariate (diagonals) and bivariate (off-diagonals) REML model.

(DOCX)

Table S5 Genetic (upper triangle) and residual (lower triangle) correlations among unrelated individuals in the ARIC population based on simultaneous analysis of all MetS traits. Mean and standard error of the Pearson correlation coefficient for genetic correlations (upper triangle) and residual correlations (lower triangle). An asterisk indicates significance with p<0.05 adjusted for 21 hypotheses using the two-tailed hypothesis test and normal distribution of the Fisher transformed correlation coefficient. (DOCX)

Table S6 Genetic (upper triangle) and residual (lower triangle) covariances among unrelated individuals in the ARIC population based on simultaneous analysis of all MetS traits. Mean and standard error. (DOCX)

Acknowledgments

We thank Peter Visscher and Marc Reitman for insightful comments.

Author Contributions

Conceived and designed the experiments: SV CCC. Analyzed the data: SV CCC. Contributed reagents/materials/analysis tools: SV JG CCC. Wrote the paper: SV CCC.

- Levy D, Ehret GB, Rice K, Verwoert GC, Launer LJ, et al. (2009) Genomewide association study of blood pressure and hypertension. Nat Genet 41: 677–687.
- Speliotes EK, Willer CJ, Berndt SI, Monda KL, Thorleifsson G, et al. (2010) Association analyses of 249,796 individuals reveal 18 new loci associated with body mass index. Nat Genet 42: 937–948.
- Teslovich TM, Musunuru K, Smith AV, Edmondson AC, Stylianou IM, et al. (2010) Biological, clinical and population relevance of 95 loci for blood lipids. Nature 466: 707–713.
- Deary IJ, Yang J, Davies G, Harris SE, Tenesa A, et al. (2012) Genetic contributions to stability and change in intelligence from childhood to old age. Nature.
- Lee SH, Wray NR, Goddard ME, Visscher PM (2011) Estimating missing heritability for disease from genome-wide association studies. Am J Hum Genet 88: 294–305.
- Czernichow S, Kengne AP, Stamatakis E, Hamer M, Batty GD (2011) Body mass index, waist circumference and waist-hip ratio: which is the better discriminator of cardiovascular disease mortality risk?: evidence from an individual-participant meta-analysis of 82 864 participants from nine cohort studies. Obes Rev 12: 680–687.
- Rice T, Province M, Perusse L, Bouchard C, Rao DC (1994) Cross-trait familial resemblance for body fat and blood pressure: familial correlations in the Quebec Family Study. Am J Hum Genet 55: 1019–1029.
- Perusse L, Rice T, Despres JP, Bergeron J, Province MA, et al. (1997) Familial resemblance of plasma lipids, lipoproteins and postheparin lipoprotein and hepatic lipases in the HERITAGE Family Study. Arterioscler Thromb Vasc Biol 17: 3263–3269.
- Frayn KN (2010) Metabolic Regulation: A Human Perspective (Frayn, Metabolic Regulation) Wiley-Blackwell. 384 p.
- Martin LJ, North KE, Dyer T, Blangero J, Comuzzie AG, et al. (2003) Phenotypic, genetic, and genome-wide structure in the metabolic syndrome. BMC Genet 4 Suppl 1: S95.

- 24. Cheverud JM (1988) A comparison of genetic and phenotypic correlations. Evolution. pp 958–968.
- Cheverud JM (1982) Relationships among ontogenetic, static, and evolutionary allometry. Am J Phys Anthropol 59: 139–149.
- Roff DA (1995) The estimation of genetic correlations from phenotypic correlations: a test of Cheverud's conjecture. Heredity 74: 481–490.
- Lander ES (2011) Initial impact of the sequencing of the human genome. Nature 470: 187–197.
- Govindaraju DR, Cupples LA, Kannel WB, O'Donnell CJ, Atwood LD, et al. (2008) Genetics of the Framingham Heart Study population. Adv Genet 62: 33–65.
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, et al. (2007) PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet 81: 559–575.
- Price AL, Helgason A, Thorleifsson G, McCarroll SA, Kong A, et al. (2011) Single-tissue and cross-tissue heritability of gene expression via identity-bydescent in related or unrelated individuals. PLoS Genet 7: e1001317. doi:10.1371/journal.pgen.1001317.
- 31. Yang J, Lee SH, Goddard ME, Visscher PM (2010) GCTA: a tool for genomewide complex trait analysis. The American Journal of Human Genetics.